Author's response to reviews

Title: Student centered curricular elements are associated with a healthier educational environment and lower depressive symptoms in medical students.

Authors:

Eiad Abdelmohsen AlFaris (efarisx2@yahoo.co.uk)
Naghma Naeem (naghma.naeem@gmail.com)
Farhana Irfan (masoodfarhana1969@gmail.com)
Riaz Qureshi Qureshi (riaz.qureshi9@gmail.com)
Cees van der Vleuten (c.vandervleuten@maastrichtuniversity.nl)

Version: 7
Date: 28 August 2014

Author's response to reviews:

Dear Editor,

The required sections (Acknowledgement, Conflict of Interest, Ethics Statement, Authors Contribution) have been added to the revised manuscript (MS ID: 1050445339106753) uploaded today 28/08/2014 on the website.

Hope it now meets all requirements.

Regards.
Dr. Naghma Naeem, MBBS, MMed, PhD.
Associate Professor & Head of Department,
Medical Education,
Batterjee Medical College.
P.O. Box 6231, Jeddah 21442- Saudi Arabia
T: +966 2 2649600, 6561111 Ext: 5109
E: naghma.naeem@gmail.com
Mededu1.jed@bmc.edu.sa